

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Thube, Sneha; Peterson, Sonja; Nachtigall, Daniel; Ellis, Jane

Article — Published Version The economic and environment benefits from international co-ordination on carbon pricing: a review of economic modelling studies

Environmental Research Letters

Provided in Cooperation with: Kiel Institute for the World Economy – Leibniz Center for Research on Global Economic Challenges

Suggested Citation: Thube, Sneha; Peterson, Sonja; Nachtigall, Daniel; Ellis, Jane (2021) : The economic and environment benefits from international co-ordination on carbon pricing: a review of economic modelling studies, Environmental Research Letters, ISSN 1748-9326, IOP Publishing, Bristol, Vol. 16, https://doi.org/10.1088/1748.0326/ac2b61

https://doi.org/10.1088/1748-9326/ac2b61

This Version is available at: https://hdl.handle.net/10419/248648

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ENVIRONMENTAL RESEARCH LETTERS

TOPICAL REVIEW • OPEN ACCESS

The economic and environment benefits from international co-ordination on carbon pricing: a review of economic modelling studies *

To cite this article: Sneha Thube et al 2021 Environ. Res. Lett. 16 113002

View the article online for updates and enhancements.

You may also like

- <u>Health co-benefits of sub-national</u> renewable energy policy in the US Emil G Dimanchev, Sergey Paltsev, Mei Yuan et al.
- Will the use of a carbon tax for revenue generation produce an incentive to continue carbon emissions? Rong Wang, Juan Moreno-Cruz and Ken Caldeira
- <u>The role of capital costs in decarbonizing</u> <u>the electricity sector</u> Lion Hirth and Jan Christoph Steckel

TOPICAL REVIEW

ENVIRONMENTAL RESEARCH LETTERS

CrossMark

OPEN ACCESS

RECEIVED 22 March 2021

REVISED 27 September 2021

ACCEPTED FOR PUBLICATION 29 September 2021

PUBLISHED 20 October 2021

Original content from this work may be used under the terms of the Creative Commons Attribution 4.0 licence.

Any further distribution of this work must maintain attribution to the author(s) and the title of the work, journal citation and DOI.



The economic and environment benefits from international co-ordination on carbon pricing: a review of economic modelling studies *

Sneha Thube¹, Sonja Peterson¹, Daniel Nachtigall² and Jane Ellis²

¹ Global Commons and Climate Policy, Kiel Institute for the World Economy (IfW), Kiel, Germany
² Environment Directorate, OECD, Environment Directorate, Paris, France

E-mail: sneha.thube@ifw-kiel.de

Keywords: climate change mitigation, harmonizing carbon pricing, fossil fuel subsidy reforms, border carbon adjustment, greenhouse gas mitigation, sectoral agreements, climate-economy-modelling

Abstract

This paper reviews quantitative estimates of the economic and environmental benefits from different forms of international co-ordination on carbon pricing based on economic modelling studies. Forms of international co-ordination include: harmonising carbon prices (e.g. through linking carbon markets), extending the coverage of pricing schemes, phasing out fossil fuel subsidies, developing international sectoral agreements, and establishing co-ordination mechanisms to mitigate carbon leakage. All forms of international co-operation on carbon pricing could deliver benefits, both economic (e.g. lower mitigation costs) and environmental (e.g. reducing greenhouse gas (GHG) emissions and carbon leakage). There is scope to considerably increase the coverage of carbon pricing, since until 2021 only around 40% of energy-related CO₂ emissions in 44 OECD and G20 countries face a carbon price. There is also significant scope to improve international co-ordination on carbon pricing: moving from unilateral carbon prices to a globally harmonized carbon price to reach the 1st round of NDC targets for 2030 can reduce global mitigation cost on average by two thirds or \$229 billion. Benefits tend to be higher with broader participation of countries, broader coverage of emissions and sectors and, more ambitious policy goals. Extending carbon pricing to non-CO₂ GHG could reduce global mitigation costs by up to 48%. Absolute cost savings from harmonized carbon prices increase by almost 70% in 2030 for reductions in line with the 2 °C target. Most, but not all, countries gain economic benefits from international co-operation, and these benefits vary significantly across countries and regions. Complementary measures outside co-operation on carbon pricing (e.g. technology transfers) could potentially ensure that co-operation provides economic benefits for all countries.

This paper is based on the OECD Environment Working Paper No. 173 (https://doi.org/10.1787/d4d3e59e-en). The views expressed here are those of the authors and do not represent the official views of the OECD or of its member countries. The authors are grateful for the comments and suggestions from three anonymous referees as well as from participants of the 5th Strategic Dialogue and the working group meetings of the Carbon Market Platform. We would also like to thank Christoph Klaiber for excellent research support. Funding for this paper is received from the German Federal Ministry of Education and Research (BMBF) within the framework of the Strategy Research for Sustainability (FONA) www.fona.de/en. BMBF funding code: CarPri (01LA1816A). The publication of this article was funded by the Open Access Fund of the Leibniz Association.

1. Introduction

Global climate action needs to increase substantially to limit global warming to 'well-below 2 °C above pre-industrial levels and pursuing efforts to limit the temperature increase to 1.5 °C above preindustrial levels' as per the target of the Paris Agreement (UNFCCC 2015). Yet, the aggregate emission reductions associated with countries' initial unconditional Nationally Determined Contributions (NDCs) would imply a 66% chance to only limit warming to 3.2 °C by the end of the century (UNEP 2019). The NDC updates that several countries have made by mid-2021 are still expected to lead to global warming of more than 2 °C (CAT 2021) though (Höhne et al 2021) show that globally the 2 °C target might be within reach if the national net-zero targets are implemented.

Pricing carbon dioxide (CO_2) and other greenhouse gas (GHG) emissions through emissions trading schemes (ETSs) or taxes is a key element of an economically efficient climate strategy. It incentivises private and public actors to reduce emissions cost-effectively while spurring innovation into zero-carbon technologies. Carbon pricing has also important synergies with broader well-being goals, enhancing public health through lower levels of air pollution while generating revenues that allow for an increase in public investments or reducing distortionary taxes (OECD 2019). Yet, carbon pricing alone is not sufficient to trigger the scale and speed of the economic transformations needed to reach the temperature goals of the Paris Agreement but needs to be accompanied by complementary policies (innovation, information provision, etc) (Tvinnereim and Mehling 2018, Stiglitz 2019). Indeed, carbon pricing so far has had only limited effects on aggregate emission reductions (Green 2021).

While the number of national and sub-national carbon pricing schemes has increased from 16 to 64 between 2009 and 2021 (World Bank 2021), around 60% of energy-related CO₂ emissions in 44 OECD and G20 countries do not face a carbon price (OECD 2021a). Indeed, only 3.8% of global emissions are priced above USD 40 per ton of CO_2 —a low-end estimate for carbon prices necessary in order to meet the goals of the Paris Agreement (World Bank 2021, OECD 2021a).

International co-operation especially but not limited to harmonized carbon pricing in a broader sense, and on meeting individual countries' emissions reduction targets is expected to bring important economic (e.g. reduce climate policy costs, fiscal revenues from allowance sales), environmental benefits (e.g. reducing GHG emissions and air pollution emissions as well as carbon leakage) and political benefits (e.g. signalling a commitment to climate mitigation to domestic and foreign stakeholders) that could potentially enhance the ambition of co-operating countries (Nachtigall 2019). Combining these benefits-for example reinvesting the savings in mitigation costs into additional mitigation or energy efficiency measures-could significantly enhance global mitigation ambition. International climate agreements have explicitly enshrined mechanisms to foster international co-operation, including in Article 6 of the Paris Agreement. Yet, evidence on the economic and environmental benefits of international co-operation is scarce and scattered. Quantifying the benefits of international co-ordination especially on pricing of GHG emissions, including CO₂ and the distribution of these benefits across country groupings can help policy makers make better-informed decisions about the implications and potential forms of international co-ordination.

This review provides a comprehensive overview of the economic and environmental benefits of a variety of forms of co-operation between countries, mainly based on economic modelling studies that can provide quantitative estimates.

2. Methods

This paper synthesises estimates of the economic and environmental benefits of international co-operation based on the economic modelling literature mostly from the past 10 years. We conducted the literature search on Google Scholar and Web of Science as the main search engines due to their vast scope and easy accessibility. On a couple of occasions, we used ECO-NIS to supplement our literature search. ECONIS is the online catalogue of the ZBW-German National Library of Economics-Leibniz Information Centre for Economic, which broadly collects economic literature and includes all major economic journals and grey literature from all major institutions undertaking economic research. We applied three general criteria for selecting studies (dominantly peer-reviewed studies and some reports and working papers from OECD, IEA, and conference papers):

- (a) We only consider studies that use ex-ante policy analysis methods. This literature typically uses numerical modelling techniques, particularly integrated assessment models (IAMs) and computable general equilibrium (CGE) models (see Annex for an overview of these modelling methods) to quantify the socio-economic and/or environmental effects of climate policies³. Therefore, we focus only on studies that use either of these models.
- (b) We only consider studies that provide quantitative estimates of economic costs measured either as a carbon price, gross domestic product (GDP) changes or welfare changes (mostly in terms of Hicksean-equivalent variation—HEV).

³ Political benefits are hard, if not impossible, to quantify.

Section 3 Harmonising carbon prices	Paris Agreement, NDCs, intended NDCs	+	Integrated assessment, general equilibrium	+	Abatement cost, mitigation cost		—
Section 4 Extending the coverage of pricing schemes	Sectoral agreements, sectoral coverage	+	Integrated assessment, general equilibrium	+	Abatement cost, mitigation cost	+	Multigas mitigation
Section 5 Multilateral FFS reform	, e	+	Integrated assessment, general equilibrium	+	Abatement cost, mitigation cost	+	FFS
Section 6 International sectoral reforms	Sectoral agreements	+	Integrated assessment, general equilibrium	+	Abatement cost, mitigation cost		
Section 7 Co-ordination mechanism for mitigating carbon-leakage	Border carbon adjustment, border carbon	+	Integrated assessment, general equilibrium	+	Abatement cost, mitigation cost	+	Carbon leakage

Table 1. List of keywords used in literature search.

(c) We focus on review studies with a multi-regional or global focus and therefore exclude articles that use a single country model. This criterion is needed because the goal of our study is to synthesise economic and environmental gains of co-operation and models need to have a multiregional or fully global representation of countries to simulate co-operation between regions. Only in very few cases where sufficient multiregional evidence was missing have we included single-country studies.

In addition to these three general criteria, specific search terms were used to select studies for each of the sections (see table 1). Particularly, our study reviews five independent instruments for initiating co-ordinated and co-operative action between countries. These are harmonising carbon prices (e.g. through linking carbon markets), extending the coverage of pricing schemes, phasing out fossil fuel subsidies (FFS), developing international sectoral agreements, and establishing co-ordination mechanisms to mitigate carbon leakage.

Section 3 focuses on price harmonization and so in this section we only selected studies that report the cost estimates for the most recent emission targets i.e. the initial NDC pledges submitted by countries under the Paris Agreement. We went through the results from these searches and selected only those studies that met the three general criteria and modelled scenarios with both unilateral prices and harmonized prices. Additionally, few papers (Springer 2003) focusing on the agreements passed in accordance with the previous Conference of Parties were looked at to supplement the full scope of the global climate change debate. For sections 4–7, topic-wise literature searches were done to expand the study to include these other four co-ordination instruments.

A snowball approach followed the first step of systematic identification of studies. This step included identifying literature from the reference list of the relevant studies found via the search engines. The final tally of 59 studies included in our study is supported by the literature search and the authors' experience. Table 2 gives an overview of the number of studies considered in each section. In addition, we include two meta-analyses (Kuik *et al* 2009, Branger and Quirion 2014) and two cross-model comparison studies (Weyant *et al* 2006, Böhringer *et al* 2021a). Therefore, the papers that are included within these four meta-analyses are not separately included in our review unless they provide unique insights.

This paper is structured as follows. Sections 3 and 4 focus on carbon pricing and discuss the benefits of harmonizing carbon prices across countries and extending the scope of carbon pricing, respectively. Section 5 deals with international co-operation in phasing out FFS which act as negative carbon prices and section 6 with international sectoral agreements. Finally, section 7 discusses options to address carbon leakage if international harmonization of climate policy fails. Finally, section 8 provides a conclusion.

3. Benefits of harmonising carbon prices

International climate agreements have explicitly enshrined mechanisms to foster international cooperation, including most recently via Article 6 of the Paris Agreement. This section reviews the economic and environmental benefits of global (section 3.1) co-operation, largely focussing on, but not limited to the goals of the Paris Agreement, and the benefits of regional co-operation (section 3.2).

Section	Name	Number of studies	Publication year of latest study
3	Benefits of harmonizing carbon prices	24	2021
3.1	Global co-operation	14^{b}	2021
3.2	Regional co-operation	10^{b}	2021
4	Extending coverage of carbon pricing schemes	11	2019
4.1	Extending sectoral coverage	8	2019
4.2	Extending GHGs	3 ^{a,b}	2012
5	Multilateral FFS reform	6	2021
6	International sectoral agreements	2	2012
7	International co-ordination on mitigating carbon leakage	16	2018
7.1	Environmental effects	13 ^{a,b}	2018
7.2	Economic effects	13 ^{a,b}	2018
7.3	Strategic incentives to join climate coalitions	3	2016
	Total	59	

Table 2.	Publication	s included	in this	paper.
----------	-------------	------------	---------	--------

Note: The superscripts a and b indicate that the sections include a meta-analysis or a multi-model study, respectively. Source: Authors.

Flexibility in the location of mitigation efforts allows for increased mitigation in countries with low abatement costs and reduced mitigation in countries with high abatement costs, achieving the aggregate emission target at a lower cost. A uniform global carbon price would, in theory, ensure that the resulting emission reductions are reached with the lowest global economic cost, regardless of whether the global price is implemented through uniform national carbon taxes (and transfer mechanism), a global ETS or full linking of national ETS (Baranzini et al 2017). Sub-global harmonisation of carbon prices could only realise some of the economic benefits. Assessing the economic and environmental benefits from harmonised carbon prices requires a comparison of achieving a specific target unilaterally (e.g. meeting NDC pledges) with achieving the same target jointly.

The aggregate cost of reaching both national and international emission reduction targets depend on four main drivers (Peterson and Weitzel 2015):

- The stringency of emission targets relative to the business-as-usual (BAU) scenario.
- The national abatement costs which are dependent on the emission intensity of production and consumption patterns, the sectoral composition of economies and technology costs.
- National and international feedback effects of climate policy through changes in relative prices of fossil energy which affect energy markets and input prices with implications on (inter)national value chains, production and consumption of other goods.
- The level of international co-operation as this could harmonise abatement costs across different sources and locations, and for some countries could also generate fiscal income from allowance trading if there are international carbon markets.

Several caveats need to be kept in mind when comparing different modelling studies.

- Different models assume different economic structures for countries and regions and make a range of different assumptions on the above-mentioned drivers.
- Quantifying mitigation pledges is not straightforward for NDCs that are not expressed as absolute emission reductions. Additional assumptions are necessary for pledges made with emissions intensity targets, emission reductions relative to prespecified baseline emissions or for different target years (2025 or 2030).
- Translating international goals related to specific temperature targets into national emission reduction targets is even more challenging in the absence of a globally agreed burden sharing agreement⁴.

Results presented here focus on aggregate results for a particular country or region; the impact for individual actors within a country or region could be significantly different from the aggregate average.

3.1. Global harmonisation of carbon prices

There is significant intra- and inter-regional variation in estimated carbon prices needed to achieve the NDCs unilaterally. Figure 1 shows the carbon prices from different models and modelling studies to achieve the NDC targets through a uniform regional carbon price compared to a global carbon price. Results diverge the most for Japan, the USA and the EU, where estimated carbon prices under unilateral action vary between USD 4–645/tCO₂-eq, USD 16–607/tCO₂-eq and USD 10–2745/tCO₂-eq, respectively. With the exception of South Africa, for the rest of the regions, the higher estimates are derived

⁴ Researchers typically analyse a number of burden sharing rules to determine the stringency of the national mitigation target for limiting global warming to 2 °C or 1.5 °C. These rules may be based on cumulative emissions, GDP, population, baseline emissions or a combination thereof (Fujimori *et al* 2016).



from models that include only energy-related CO_2 In some regions emission reductions and exclude lower-cost land-use ults suggest carl emission reductions. Yet, it should also be noted, those regions we

emission reductions. Yet, it should also be noted, the full set of 49 models, includes 44 models with only energy-related CO_2 emissions and only five that include land-use emissions.

Included studies: (Fujimori *et al* 2016, Vandyck *et al* 2016, Aldy *et al* 2016a, Aldy *et al* 2016b, Akimoto *et al* 2017, Dai *et al* 2017, Liu *et al* 2020, Böhringer *et al* 2021a).

The substantial difference in carbon prices across regions to meet a given target in all reviewed studies highlights the large potential gains from international co-operation in reducing the costs of emission reductions. Regional carbon prices tend to be highest in advanced economies (US, EU, Japan, Canada) with average carbon prices around USD 47–119/tCO₂-eq. Note that for all regions this is significantly higher than currently observed carbon prices. Also, the current EU-ETS price of above USD 50/tCO₂-eq is well below the USD 119/tCO₂-eq average price the reviewed studies find for Europe. Altogether, in OECD and G20 countries, less than 10% of GHG emissions were priced above USD 100/tCO₂-eq in 2018 (OECD 2021a).

Given this divergence between actual and modelled carbon prices, the results reported in this section should be interpreted as an upper bound of real-world effects of international co-operation. Simulated prices tend to be lowest in emerging economies (e.g. Russia, India, China and South Africa). In some regions (Russia and India), some model results suggest carbon prices to be zero, implying that those regions would reach their NDC targets under BAU. Low carbon prices could reflect the limited ambition of mitigation targets or a large potential of low-cost abatement options. Other metrics of mitigation costs (e.g. loss of GDP compared to BAU) would result in different regional orderings of costs. If NDCs were achieved jointly (e.g. through a global carbon market), the global carbon price is estimated to be between USD 0.2 and USD 58/tCO2-eq with an average of USD 18.3/tCO₂-eq. This result of requiring a lower carbon price with joint effort relative to the unilateral effort is in line with the findings for the Kyoto Protocol of 13 models reviewed in Springer (2003). They showed that the average carbon price for unilateral action in the regulated annex B countries to meet their Kyoto target was three times higher than with global trading (USD 27/tCO₂-eq vs USD 9/tCO₂-eq, respectively).

In the studies that include global cost measures and global co-operation, harmonization of carbon prices would reduce total mitigation costs relative to the unilateral achievement of NDCs. Relative to unilateral carbon pricing, 80% of the models show that harmonized carbon prices result in cost reductions (either in GDP or in terms of welfare) in the order of 48%–83% (Fujimori *et al* 2016, Akimoto *et al* 2017, IETA 2019, Böhringer *et al* 2021a) and the average is a cost reduction of 64%. This would translate into annual cost savings (see figure 3), estimated variously



from zero to USD 1240 billion in 2030⁵. Eighty per cent of the values are in the range of USD 51–365 billion.

Included studies: (Nordhaus 2015, Fujimori *et al* 2016, Qi and Weng 2016, Aldy *et al* 2016b, Akimoto *et al* 2017, Vrontisi *et al* 2018, Wei *et al* 2018, IETA 2019, Böhringer *et al* 2021a).

Going beyond achieving current NDCs jointly, co-ordination on achieving more stringent mitigation targets including those that are compatible with limiting global warming to 1.5 °C or 2 °C relative to pre-industrial levels has a number of implications. First, more ambitious mitigation targets are likely to translate-at least in the shorter term and without accounting for the benefits of climate action-into higher direct regional and global mitigation costs both in terms of necessary global carbon prices to achieve this global target (see figure 2) and of GDP/welfare loss relative to BAU. In the 2 °C scenarios, carbon prices increase by on average 4.4 times compared to the NDC scenarios with a range of 2-10.8 times. In the 1.5 °C scenarios they increase on average by five times compared to the 2 °C scenarios with a range of 1.6-5.3 times. Only Fujimori et al (2016) report carbon prices for all three climate targets and in the results the price for 1.5 °C target is 35 times that for the NDC targets and ten times that for the 2 °C target. Reported changes in GDP/welfare are of the same order. As more stringent targets

would translate into higher regional carbon prices that would further diverge, and hence, price harmonization would also increase the absolute gains of international co-ordination (IETA 2019). The model comparison study of (Böhringer et al 2021a) finds that through co-operation, the costs (measured as changes in welfare relative to unilateral action) on average reduce by 50% in 2030 for emission reductions in line with the 2 °C target and that 80% of the models report cost reductions within the range of 32%-68% reductions. The full range of costs reductions across all models is 0%–82%. In absolute terms, this translates into average welfare gains of USD 391 billion in 2030 (see also figure 3). Thus, absolute gains of coordination increase under more ambitious mitigation targets, whereas the relative gains decrease. This is also stressed by one study (IETA 2019) that analyses targets further in the future which are also more ambitious. This study, (IETA 2019) estimates absolute gains of full international co-ordination would increase from USD 249 billion in 2030 to USD 345 billion in 2050 and USD 988 billion in 2100. Relative gains would decrease from a cost reduction of 63% in 2030, 41% in 2050 and 30% in 2100.

Included studies: (Fujimori *et al* 2016, Qi and Weng 2016, Akimoto *et al* 2017, Hof *et al* 2017, IETA 2019, Böhringer *et al* 2021a).

The identified gains are not shared equally across countries. This is in particular shown by country-level results in Böhringer *et al* (2021a) where at least some of the models show that Africa, Australia/New Zealand, China, Middle East, Russia, South Korea, USA, Other Americas, and especially, Japan and India have lower welfare costs when NDCs are reached without co-operation and unilateral carbon prices than under a global carbon price. For India, this is even the case for the average across all models. Only Europe, Canada, Brazil and the rest

⁵ Akimoto *et al* (2017) do not explicitly report the cost savings from global emissions trading. However, assuming a global GDP of USD 117 trillion in 2030 (EIA 2017), the reported reduction of 0.16% in the co-ordinated case instead of 0.38% in the unilateral achievement of the NDCs would imply cost savings of around USD 259 billion. Note that both Böhringer *et al* (2021a). and Fujimori *et al* (2016) uses loss in terms of welfare measured as HEV as cost metric. The values for Böhringer *et al* (2021a) are not included in the paper but were provided by the authors.



of Asia unambiguously gain from co-operation in all models. On average, gains are most pronounced in Russia and the Middle East. These findings, are (Böhringer et al 2021a) driven especially by changes in fossil fuel prices and fossil fuel demand and also (Fujimori et al 2016) competitiveness effects on world markets. Under global co-operation, abatement shifts to the cheap reduction of coal consumption in China and India implying fewer reductions in oil and gas. This is beneficial for large oil and gas producers (Böhringer et al 2021a). Producers in countries with projected high unilateral carbon prices such as Canada and Europe that can import allowances under global co-operation significantly benefit from the lower carbon prices brought about by global cooperation on carbon pricing, since this improves their position on world-markets.

Through the same mechanism, producers in allowance-selling countries (e.g. China and India) incur higher costs despite the revenues from selling allowances. Both China and India are characterised by a carbon-intensive economic structure and low abatement costs (and carbon prices) under unilateral NDC achievement. A global carbon market would raise their carbon prices, putting a relatively large burden on their emissions-intensive economy and negatively affecting their international competitiveness vis-à-vis more developed and less emissions-intensive economies (Fujimori et al 2016). The same is also true for consumers that gain from co-operation if this decreases national carbon prices relative to unilateral action and suffer from global cooperation if it increases national carbon prices relative to unilateral action. In principle, the economic gains from trading for other countries would provide scope to make a global carbon market beneficial for all countries. This

could be done in different ways (e.g. via transfers of technology or finance), which are not further assessed here and which could vary widely in terms of political feasibility.

Besides differences in the gains from co-operation across countries, also different household-types are affected differently from carbon pricing and potentially also from co-operation. In general, the distributional effects of carbon pricing depend on the chosen ways of revenue recycling, the differences in carbon intensities of consumption across different income groups, and varying income sources (labour vs capital income) of different household-types. As laid out in Böhringer et al (2021a), carbon pricing without revenue recycling is typically regressive-hurting lower income groups that spend a larger share of their income for energy relatively more than richer households. Revenue recycling e.g. through lumpsum transfers to households can still lead to overall progressive impacts (Böhringer et al 2021a). Unfortunately, we did not identify studies that analysed the distributive effects of co-operation on within country burden sharing.

3.2. Regional harmonisation of carbon prices

Regional harmonisation of carbon prices would reduce mitigation costs of the regional coalition, but to a lower extent than the reduction under full global co-operation. Regional harmonisation could be achieved through linking existing or prospective ETSs which will achieve a uniform price in all regions or through minimum carbon prices as in Canada under the Pan-Canadian Framework on Clean Growth and Climate Change for climate change which will at least reduce the price gap and the resulting inefficiencies. All of the 14 studies we review include the EU. Six of the studies including one multi-model study assess an EU ETS-China linkage, three studies analyse a link between the EU and different coalitions of countries, including G20 countries (e.g. Canada, Japan, Russia, Australia, India, Brazil) and six of the studies cover multi-regional linkages (e.g. annex I countries⁶). The studies evaluate different reduction targets, extent of sectoral coverage in countries involved and timing and extent (unrestricted versus restricted) of linking, making it difficult to compare these studies. Nevertheless, some common points can be identified.

Studies show that not all countries would gain from linking compared to not linking. The countryspecific economic benefits from linking would depend strongly on the country's marginal abatement cost, assumed reduction targets and whether the country is an exporter or importer of emission allowances. In most studies, developed countries are assumed to have the strictest emissions mitigation targets and, thus, the highest carbon prices pre-linking. Linking with jurisdictions with lower carbon prices would reduce the allowance price, leading to benefits in most cases. For instance, in the EU ETS-China studies (Liu and Wei 2016) find that mitigation costs could be reduced by as much as 66% compared to not linking, notably when the price difference pre-linking was very high as do most of the other studies. Conversely, allowance-selling countries would not always have economic benefits from linking compared to no-linking as such countries would be negatively affected by rising carbon prices (Hübler et al 2014, Gavard et al 2016, Böhringer et al 2021a) and thus, require compensation. The aggregate gains compared to no-linking would be lower if linking was restricted as in Li et al (2019). Region-specific results include:

- Australia is expected to be a buyer of allowances in all analysed scenarios and would gain in terms of welfare in all scenarios (Böhringer *et al* 2014a).
- The EU would be buying allowances and gaining in terms of welfare (with the exception of an EU— Australia ETS (Nong and Siriwardana 2018) or an ETS that covers all annex I regions (Dellink *et al* 2014).
- China is found to be a seller of allowances in all studies, but would not benefit from linking in some studies relative to unilateral achievement of mitigation targets in the absence of additional transfers (Gavard *et al* 2011, Böhringer *et al* 2021a) or raised climate ambition of linking partners (Liu and Wei 2016).
- In a linked Asian ETS covering China, South Korea and Japan set-up to jointly reach the NDC targets, induces gains mainly for South Korea, while all 15

models of the cross-model comparison only report minor changes in adjustment costs for China and Japan (Böhringer *et al* 2021a).

• For Canada, Japan and the US, there is no clear conclusion.

Extending the geographical scope of carbon markets would reduce the aggregate mitigation costs of participating countries but would again not benefit all countries. Adding new coalition members could increase or decrease the allowance price of the extended coalition, depending on the carbon price associated with the new member(s). If the allowance price increased, former allowance importing regions would likely experience a decrease in welfare compared to the status quo in the absence of additional transfer payments as they need to pay higher prices to offset their emission obligations. For example (Gavard et al 2016) find that if the EU or the US joined a US-China or EU-China coalition, the mitigation costs of the existing coalition members would increase whereas those of the new member would decrease. Conversely, in Alexeeva and Anger (2016) allowance importing countries tend to gain if the entrance of new countries in the coalition reduces the allowance price. Also Böhringer et al (2014a) find that if the allowance price decreases with the extension of the existing coalition, allowance-selling countries may not benefit relative to no extension.

4. Extending coverage of carbon pricing schemes

Energy-related CO₂ emissions from electricity and energy-intensive sectors represent the largest share of emissions covered by existing carbon pricing schemes although some large schemes also include other emissions sources (ICAP 2019). This means that current carbon pricing schemes exclude a number of low-cost abatement opportunities in other sectors (e.g. buildings, agriculture) or from non-CO₂ (NC) GHGs (e.g. methane, nitrous oxide and F-gases), which are not always included in the models reviewed in the previous section. NC-GHGs differ from CO₂ both in terms of radiative efficiency and atmospheric lifespan, making it challenging to calculate a standardised metric. UNFCCC (and the reviewed models) use the global warming potentials over 100 years, but this metric does not adequately capture different behaviours of short-lived (e.g. methane) versus long-lived (e.g. CO₂) climate pollutants (Cain et al 2019).

4.1. Extending sectoral coverage of pricing schemes Expanding sectoral coverage would generally reduce aggregated mitigation costs through harmonising carbon prices across sectors (Böhringer *et al* 2009, 2014a, Mu *et al* 2018) while also reducing the risk

⁶ Annex I countries include most developed economies. For a list, see: www.oecd.org/env/cc/listofannexicountries.htm.

of inter-sectoral leakage⁷ (Söder *et al* 2019). The benefits from expanding sectoral coverage are higher the greater the risk of inter-sectoral leakage and the higher the difference of marginal abatement costs before the extension.

Böhringer et al (2014a) show that step-wise expanding sectoral coverage (e.g. beyond electricity and energy-intensive industry) of hypothetical international carbon markets would reduce mitigation costs for the vast majority of countries. This study also finds that international emissions trading covering only the power sector yields the highest cost savings. They find that a hypothetical link between an EU and US ETS covering only the power sector would reduce aggregate mitigation costs by around 14% by 2020 compared to the unilateral achievement of targets. Expanding the coverage to other sectors (e.g. energy intensive industry, road transport, aviation, all industrial sectors) from the EU–US power market link could further reduce mitigation costs by up to 4% points. This pattern of results also holds true for other combinations of countries, beyond an EU-US link.

The multi-model study by (Böhringer *et al* 2021a) also includes a scenario with a global ETS covering all sectors versus a scenario where only the energy and trade exposed (EITE) sectors plus the power sector are covered. Global gains from such a partial ETS relative to no co-operation in the reported NDC scenario are still positive in all models but average gains are reduced by around a 3rd. In a study for China, Mu et al (2018) find that real GDP in 2030 is reduced by 2.1% relative to a no policy scenario if China reaches its NDC through an economy-wide ETS. This GDP reduction relative to the no-policy case increases to 10.5% if the ETS only covers eight energy intensive sectors (petrochemicals, chemicals, construction materials, iron and steel, non-ferrous metals, paper, electricity, and air transport) that were responsible for 52% of Chinese CO₂-emissions in 2012. With an ETS that adds nine additional energy intensive sectors so that the ETS covers 76% of 2012 CO2emissions, real GDP reduces by only 3.3% relative to a no policy case. Thus, the analysed sectoral expansion reduces costs by almost a 3rd. The reviewed studies on specifically extending the coverage of the existing EU ETS to the transport sector find that this would enhance economic efficiency (Abrell 2010, Flachsland et al 2011, ECF 2014, Heinrichs et al 2014). In all these studies, the transport sector would be an allowance buyer. Including transportation into the EU ETS could lower mitigation costs compared to a scenario in which transport is excluded from the EU ETS, but faces additional (e.g. on top of existing gasoline taxes) carbon prices to reduce transport emissions. Yet, the

result of (Abrell 2010) is that a reallocation of mitigation obligations from transport to the sectors currently covered by the EU ETS would reduce mitigation costs even more than including transport into the EU ETS⁸.

4.2. Extending coverage of pricing schemes to NC-GHG emissions

The abatement potential of NC-GHG emissions is large and predominantly originates from the landuse, land-use change and forestry (LULUCF) sector, but also the energy sector (e.g. methane from natural gas extraction and transmission) (IPCC 2014). Some ETS cover multiple gases, but only a few (e.g. New Zealand) are currently planning to price emissions and removals from the LULUCF sector (ICAP 2019).

Extending the coverage of pricing schemes towards NC-GHGs in all economic sectors would reduce mitigation costs as shown in figure 1. A crossmodel comparison (Weyant et al 2006) of 19 global energy models simulate a least-cost policy scenario that is in line with stabilising radiative forcing at $4.5 \text{ W} \text{ m}^{-2}$ relative to pre-industrial times by the year 2150⁹. Their results show that in the 21st century, carbon (equivalent) prices in the multi-gas scenario would be, on average, between 23% and 48% lower than carbon prices in the CO₂-only scenario (Weyant et al 2006). This result holds for all but one model in this study. At the same time, the global GDP losses with multi-gas mitigation are between 0.1% and 4.8%, while those with only CO₂ mitigation range between 0.1% and 6.4%. The maximum difference in cost reduction of 0.3% points by 2025 when including NC-GHGs would amount to annual savings of USD 197 billion, almost equivalent to the reported size of global savings in mitigation costs from unrestricted emission trading to reach the NDCs (see section 3.1).

The general results are confirmed by two other studies. Ghosh *et al* (2012) provide an analysis of CO_2 mitigation policies versus all GHG mitigation policies and generally, extending carbon pricing coverage to include NC-GHGs would also reduce mitigation costs in terms of GDP loss compared to BAU. Ghosh *et al* (2012) also find that a uniform price on global GHG emissions would unambiguously benefit all countries or regions due to the gain in flexibility. The 2nd study

⁷ Inter-sectoral leakage refers to a situation, in which a sectorspecific climate policy leads to an increase of emissions in a nonregulated sector in the same country.

⁸ The reason is that constraining transport emissions substantially would reduce tax revenues from pre-existing fuel taxes, leading to a negative welfare effect (Abrell 2010). Yet, this study does not account for other externalities of road transport, including congestion, accidents, and health impacts due to noise, which tends to be larger than the social cost of carbon. Reallocating mitigation obligations from road transport to other sectors would lead to an increase in traffic, exacerbating the negative costs and potentially outweighing the tax interaction effect.

 $^{^9}$ The representative concentration pathway (RCP) 4.5 is not compatible with the Paris Agreement as it is more likely than not to result in global temperature rise between 2 °C and 3 °C relative to pre-industrial levels (IPCC 2014).

IOP Publishing

is a meta-analysis based on 26 models by Kuik *et al* (2009) and also includes results from Weyant *et al* (2006). They conduct a meta-regression analysis and estimate that the MAC estimates in 2025 are lower by 48% and by 40% in 2050 with multi-gas mitigation rather than CO_2 -only mitigation (in line with the results from Weyant *et al* (2006)).

5. Multilateral FFS reforms

FFS result in artificially low energy prices, encouraging carbon-intensive modes of consumption and production. In 2009, G-20 leaders called countries to 'rationalise and phase-out inefficient FFS that encourage wasteful consumption over the medium term' (G-20 2009). Decreasing international oil prices, the FFS reform momentum, as well as international peer-reviews of national FFS (e.g. Canada, China, Germany, Mexico, US) led to a reduction of FFS between 2013 and 2016 in 76 countries (IEA and OECD 2019). However, estimates also show that in 2017, FFS increased by 5%, reaching USD 340 billion. Despite the pledges of G20-leaders in 2009, FFS in 2020 are still at the same level as in 2010 (OECD 2021b). Consumer FFS account for around 75% of FFS in OECD and partner countries. In Jewell et al (2018), a global phase out of FFS by 2030 is estimated to reduce global CO₂ emissions by 1%-4% relative to BAU. Previous studies indicated that a global phase out of FFS by 2020 could reduce global CO₂ emissions by 5% to 6% by 2035 (Schwanitz et al 2014) and 6%-8% by 2050 compared to BAU (Burniaux and Chateau 2014). A more recent analysis by Chepeliev and Dominique (2020) shows that depending on the oil prices, removal of consumption FFS could reduce global emissions by 1.8%-3.2% in 2030. Figure 4 provides a range of global and country-specific emission reductions in response to a global FFS phase-out.

The reviewed studies show that phase out of consumer FFS would reduce emissions in reforming countries, increasing emissions elsewhere, leading to carbon leakage. For example, Burniaux and Chateau (2014) find that FFS removal in non-OECD countries would reduce global CO2 (and GHG) emissions by 10% compared to BAU. However, while CO2 emissions in non-OECD countries would decrease by 16%, emissions in OECD countries would increase by 7% compared to BAU by 2050. All relevant studies find that emission reductions in 2050 with FFS reform tend to be largest in fossil fuel exporting countries, including Russia and Middle Eastern and North African countries, amounting to 45% (Burniaux and Chateau 2014), 20% (Schwanitz et al 2014) and 2%–10% (Jewell et al 2018). Lower energy demand in energy exporting countries would translate into reduced global energy prices, which could increase fossil fuel consumption and emissions in energy importing countries (e.g. Europe and Japan). Due to this so-called 'energy price channel', carbon

leakage could also arise in case of a global phase out of FFS (Jewell *et al* 2018).

Sub-global phase out of FFS is less effective than global phase out. If only G20 countries removed FFS by 2020 ('G20 scenario'), then global GHG emissions would reduce by merely 1% by 2050 compared to BAU (Schwanitz et al 2014). This number would rise to almost 3%, half the reduction of a global phase out, if in addition to the G20 countries all member countries of the Asia-Pacific Economic Co-operation (APEC) removed their FFS (Schwanitz et al 2014). Carbon leakage, notably to Europe, the US and Japan, would be lower for smaller coalitions of reforming countries. For example, Japan's GHG emissions would hardly be affected by a phase out of FFS in G20 countries only, but would increase by 3%-7% for phase outs of G20 + APEC and global phase out, respectively (Schwanitz et al 2014). This pattern is seen because the repercussions of FFS reform on international energy prices are lower for smaller coalitions. While in the G20 scenario, international oil prices would drop by 2% and international gas prices would be hardly affected at all, those prices would decrease by 5%-10%, respectively, under a global phase out (Schwanitz et al 2014).

Böhringer *et al* (2021b) show that phasing out producer FFS could lead to negative carbon leakage rates, i.e. decreased emissions in countries not phasing out FFS. Removing producer subsidies (i.e. transfers from taxpayers to producers of fossil fuels) leads to an increase in producer's production costs and, thus, increases both the domestic and international price for fossil fuels, reducing demand emissions both domestically and abroad.

Included studies: (Burniaux and Chateau 2014, Magné *et al* 2014, Schwanitz *et al* 2014, Jewell *et al* 2018, Chepeliev and Dominique 2020).

All studies assessed for this paper indicate that joint global welfare would increase with a coordinated FFS reform. Moreover, Schwanitz et al (2014) finds that the gains in aggregate welfare would increase with an increasing number of co-operating countries and, thus, in the size of FFS removals. Burniaux and Chateau (2014) find that removing consumer subsidies in non-OECD countries could lead to a 5% welfare increase (due to lower energy prices) in OECD economies, but only to a 0.2% welfare increase in non-OECD countries. They also find that some countries (e.g. Russia) may not benefit from co-ordinated FFS removal in the absence of additional transfers. Chepeliev and Dominique (2020) find that the total removal of all FFS would increase global welfare between 0.02% and 0.1% in 2030 relative to BAU, depending on the oil prices. Similar to Burniaux and Chateau (2014), Russia also faces welfare losses in Chepeliev and Dominique (2020). The results on welfare and emissions are summarised in figure 4. Unilateral FFS phase-outs frees up public budget spent on FFS, that could be invested



for other purposes or allocated to households, and could trigger a more efficient domestic allocation of resources, both of which would generally enhance domestic welfare. Burniaux and Chateau (2014) find that under unilateral phase out, energy exporting countries would see the largest welfare gains by 2050 compared to BAU (4%), followed by India (2.3%) and China and Russia (0.4%). In contrast, multilateral phase out of all non-OECD countries would alter the distribution of welfare gains and losses to 2050: Russia would face a welfare loss of 5.8%, oil-exporting countries would show no change in welfare and India and China would gain by 3.0% and 0.7% compared to BAU, respectively. The reason is that a multilateral FFS removal would lead to a large decrease in energy demand and global energy prices, reducing the value of fossil fuel exports for energy exporters and offsetting the initial efficiency gains from the reform.

6. International sectoral agreements

Sectoral agreements could be one avenue through which (international) carbon prices could be implemented or harmonised for specific economic sectors. Such agreements have the potential to reduce sectorspecific GHG emissions while addressing concerns on competitiveness and carbon leakage in industrialised countries, as well as on economic development in emerging countries (Meunier and Ponssard 2012). Bottom-up sectoral approaches could set binding, but potentially regionally differentiated emission targets for specific sectors, including aviation and EITE sectors. Current sectoral approaches include the Carbon Offsetting and Reduction Scheme for International Aviation, aiming to stabilise global international aviation emissions at 2019 levels (ICAO 2020), and pledges of the International Maritime Organisation (IMO) to reduce GHG emissions from international shipping by at least 50% by 2050 compared to 2008 'whilst pursuing efforts to phase them out' (IMO 2018).

Results from modelling the impact of sectoral agreements on GHG emissions are limited, but suggest that such agreements could reduce GHG emissions although not cost-effectively. The conclusion is based on only two studies for the cement sector (Voigt et al 2012) and the energy-intensive sectors (Akimoto et al 2008)¹⁰. Sectoral approaches could reduce GHG emissions in industrialised, emerging and developing countries regardless of whether they stipulate absolute (Voigt et al 2012) or emission intensity (Akimoto et al 2008) targets in the sectors covered. These agreements would also mitigate competitiveness concerns of sectors and could increase the welfare of participating countries compared to unilateral achievement of sectoral mitigation targets¹¹. However, compared to policy scenarios with a uniform global carbon

 $^{^{10}}$ Other studies are exploring the technical potential and the costeffectiveness of international co-operation in the low carbon transition of specific sectors, including cement (Cembureau 2013) or iron and steel (WSA 2019). Since they lack international and crosssectoral repercussions they are not further discussed here. 11 For example, Voigt *et al* (2012) find that the decrease of EU

price, sectoral approaches would incur larger welfare losses.

7. International co-ordination on mitigating carbon leakage

Climate policy that is not internationally harmonized faces the risk that economic activities and their associated emissions relocate from countries with higher carbon prices or stricter climate policy to countries with lower prices or less stringent climate policy. This is referred to as carbon leakage and denotes a situation where the benefits of emissions reduction in a given location are partially offset by emissions increases elsewhere. Co-ordinated regional implementation of carbon pricing, e.g. through carbon clubs (IISD 2018) or climate coalitions (see section 3.2) could reduce carbon leakage within the coalition, but could increase the risk of carbon leakage outside. Furthermore, as economic activity may relocate to countries with lower carbon prices, this would lead to welfare losses in the implementing countries, including loss of jobs and tax revenues, while undermining the environmental effectiveness of carbon pricing.

In the absence of deeper international co-operation, regional or unilateral anti-leakage policies could address carbon leakage but these are always second best to co-ordinated international climate policies. Anti-leakage policies could increase the environmental effectiveness of international co-operation on carbon pricing by ensuring that emission reductions in a climate-coalition are not offset by emissions increases outside the coalition. As such, anti-leakage policies could, enhance economic outcomes (for coalition members), and/or incentivise more international co-operation.

Most existing carbon pricing schemes address carbon leakage through preferential tax rates, fuel excise taxes or free allocation of emission allowances for ETS, notably for the energy-intensive trade-exposed industry which are most affected by differences in international carbon prices (Ellis *et al* 2019).

Border carbon adjustments (BCAs) was recently also proposed as part of the European Green Deal package. BCA have a number of practical (e.g. measurement of the carbon content), legal (e.g. WTO compatibility) and political challenges (e.g. feasibility, risk of amplifying retaliation measures), which need to be weighed against the potential benefits (Cosbey *et al* 2019). Our interest here is in how far they can address the carbon leakage problem.

7.1. Effects of anti-leakage policies on GHG emissions

A meta-analysis by Branger and Quirion (2014) that reviewed 25 modelling studies shows that in the absence of any anti-leakage policy the leakage rates of regional or unilateral climate policy is estimated to range between 5% and $20\%^{12}$. This contrasts to the empirical ex-post literature, which does not find any evidence of carbon prices on carbon leakage (Dechezleprêtre *et al* 2019, Naegele and Zaklan 2019, Venmans *et al* 2020). Most of the ex-post studies also do not find negative and statistically significant effects of carbon pricing on firms' competitiveness (Venmans *et al* 2020). Part of the reason is that actual carbon prices have been low and safeguards for the industry were in place (e.g. free allowances).

In the modelling literature, the leakage rate depends on a number of factors:

- More stringent mitigation targets would result in higher leakage rates (Böhringer *et al* 2012b, Branger and Quirion 2014). The reason is that more stringent mitigation targets would imply higher implicit carbon prices, leaving more scope for carbon leakage. In view of the ambition needed to achieve the goals of the Paris Agreement, this finding highlights the importance of international co-operation to enhance environmental effectiveness and mitigate carbon leakage. Larger coverage of GHGs would decrease carbon leakage (see section 4.2) due to increased flexibility of meeting abatement targets.
- Increasing coalition size would reduce the leakage rate (Böhringer *et al* 2012a, 2014a, Branger and Quirion 2014). Böhringer *et al* (2014a) systematically assess the effects of coalition size on different anti-leakage measures and report that the differences in leakage rates between anti-leakage instruments reduce with increasing coalition size.
- Harmonising the carbon price within the climate coalition would tend to reduce the leakage rate. This is because a harmonised price minimises the trade repercussions in global energy markets. Model assumptions and choices also have a large influence on estimated leakage rates. First, carbon leakage estimates are higher in CGE models than in partial equilibrium models because the former explicitly includes international repercussions affecting the leakage rate. Second, higher trade elasticities (i.e. fewer trade frictions) increase

covered by the EU ETS) would be 9% lower when emerging economies (China, Brazil, Mexico) also adopt sector-specific mitigation targets in that sector. Including these countries' cement sectors in the EU ETS so that all cement facilities face the same carbon price would reduce the loss in EU cement output by even 36%. If the cement sector in all countries faced the same carbon price, this would reduce the welfare loss in the EU and China relative to a climate scenario without sectoral targets while only slightly lessening the welfare of Mexico, leaving Brazil's welfare unchanged. In contrast, unilateral sectoral targets would lessen the welfare in all emerging economies.

 $^{^{12}}$ A leakage rate of 5% implies that a climate policy leading to a reduction of 100 CO₂e emissions within the climate coalition would increase emissions by five CO₂e in countries outside.

leakage, allowing price shocks to transmit more heavily in international energy markets. This finding is strengthened through Böhringer *et al* (2017) that includes scenarios with different trade elasticities.

Studies that compare different anti-leakage instruments find that all of them would reduce the risk of leakage, but BCA are expected to be the most effective instrument. BCA would lead to the lowest leakage rate compared to free allocation of allowances and industry tax exemptions for different coalitions and different emission reduction targets (Böhringer et al 2010, 2012a, 2012b, Monjon and Quirion 2011). Yet, no anti-leakage policy could entirely mitigate leakage. Branger and Quirion (2014) in their metastudy find that BCA would reduce the leakage rate on average by 6% points compared to scenarios where emission reduction targets are reached without BCA. The reduction in the leakage rate is estimated to be between 1% and 15% points with some outliers as high as 30% points. More recent studies (Antimiani et al 2016, Böhringer et al 2017, 2018, Larch and Wanner 2017) also report results within the range of the meta-study and in most studies, none of the anti-leakage policies would be able to completely offset leakage. This is because these policies only target the trade channel but do not explicitly address the energy channel. Hence, Burniaux et al (2013) find that BCA would be more effective in reducing leakage for rather small coalitions that have less influence on global fossil fuel prices13. Böhringer et al (2017) also stress that the negative leakage rate they find for BCA stems from the fact that energy market effects are not considered here.

Most relevant for this paper is the finding that larger coalitions would lead to a lower leakage rate while broadening the regional coverage of GHG emissions, making climate policy more effective. In fact, the size of the coalition of co-operating countries is the single most important factor that determines the extent of carbon leakage (Branger and Quirion 2014). This also highlights the importance of international cooperation as a first-best policy before turning to antileakage instruments. As the coalition size increases, the number of regions where emissions could leak to decrease (to zero, in the case of a global coalition with a uniform carbon price). The results from the meta study suggest, on average, a 37% reduction of the leakage rate if instead of only European countries, all annex I countries except Russia reduced their CO2 emissions by 15% relative to a benchmark (Branger and Quirion 2014). In some studies, reduction of leakage rates for the same regional extension of the coalition could be as high as 60% (Ghosh *et al* 2012, Böhringer *et al* 2012a). Adding China to the coalition would reduce the leakage rate by an additional 50% (Ghosh *et al* 2012).

7.2. Economic and welfare effects of anti-leakage instruments

BCA would be expected to be beneficial for the coalition countries. The results of Branger and Quirion (2014)'s meta-study suggest that the change in welfare (not accounting for the welfare effect from emission abatement) compared to BAU in the abating coalitions would range from -1.6% to -0.02% without BCA and from only -0.9% to +0.4% with BCA. Hence, BCA would reduce coalition countries' welfare loss by up to 44%. One of the drivers is that BCA tend to mitigate the reduction in output from climate policy in EITE sectors (Böhringer *et al* 2014b). Yet, BCA would usually not be able to restore the welfare levels of BAU scenarios (i.e. without climate policy) since coalition countries still face direct abatement costs¹⁴.

In many but not all studies BCAs reduce negative welfare effects of unilateral climate policy in the model-regions undertaking this climate policy but mostly they do not establish a cost-neutral situation in the sense that with BCAs, the model regions do not reach the same level of welfare as without any climate policy. BCA would transfer part of the mitigation burden to the non-coalition countries whose exports are taxed (Walley 2012, Burniaux et al 2013, Böhringer et al 2014b, Larch and Wanner 2017, Böhringer et al 2018). Energy-exporting countries would typically incur the largest welfare loss due to BCA (Weitzel et al 2012, Böhringer et al 2018). The welfare losses incurred by non-coalition countries would partly offset the welfare gains of coalition members. Yet, global welfare would decrease as a result of BCA relative to a policy scenario without BCA, also because it causes additional emission reductions (Branger and Quirion 2014).

Allocating free allowances or tax exemptions for industry transfers income from governments to industrial sectors without necessarily changing trade patterns. In contrast to BCA, this would not negatively affect non-coalition countries, but would also not benefit the coalition countries. Yet, the joint welfare loss of both country groups would be higher for allocating free allowances than for BCA (Böhringer *et al* 2017).

¹³ The results of few studies suggest that implementing BCA would even result in negative leakage rates, meaning that BCA offsets the negative competitiveness effect, and reduces emissions in noncoalition countries (Branger and Quirion 2014).

¹⁴ Few studies suggest that the welfare of coalition countries under BCA would be higher than under BAU. This surprising result can derive from trade policy effects, according to which indirect termsof-trade benefits from taxing exports of foreign countries realised by coalition countries (e.g. OECD) more than offset direct abatement cost for major industrialised regions such as Germany, the United States and Japan (Böhringer *et al* 2018).

7.3. Strategic incentives to join climate coalitions

As noted above, BCA would usually reduce the welfare of non-coalition members compared to no BCA, providing incentives for countries to avoid the negative welfare effects by joining the climate coalition Such incentives are mostly analysed using stylized and partly also parameterized game theoretic models¹⁵ and a few CGE models (Weitzel *et al* 2012, Böhringer et al 2016). Overall, they find that BCA could induce participation in climate coalitions but only under very specific assumptions (Böhringer et al 2016, Al Khourdajie and Finus 2020) or countries. BCA would entice more ambitious climate policy outside the coalition only for very low levels of climate ambition (and thus carbon prices) of the coalition (Nordhaus 2015). In fact, BCA would not be able to create a stable global climate coalition even for very low levels of carbon prices. While club participation could be 13 out of 15 model regions for carbon prices below USD 10, participation decreases to two regions for carbon prices above USD 10 (Nordhaus 2015). Energy-exporting countries tend to have the largest incentive to join the coalition as they are most adversely affected by BCA (Weitzel et al 2012, Böhringer et al 2016) while the studies find incentives only under unrealistic assumption or not at all for other countries and regions. Other hypothetical measures, notably trade tariffs would be more effective than BCA to incentivise non-coalition countries to join the coalition, but would likely breach multilateral trade rules (Nordhaus 2015)¹⁶.

8. Summary and conclusions

This paper assesses quantitative estimates of the economic and environmental benefits from different types of international co-ordination on carbon pricing based on economic modelling studies. Better awareness and understanding of these benefits could encourage governments to increase

¹⁶ The result of one study suggests that international compensating transfers in form of additional emission allowances are a more efficient instrument to create a stable global coalition than BCA, leading to larger global welfare levels (Weitzel et al 2012). Trade tariffs could also trigger participation in global climate coalitions when used against non-coalition members because tariffs would increase the cost of non-participation (Lessmann et al 2009, Nordhaus 2015). Trade tariffs of 1% (Nordhaus 2015) and 1.5% (Lessmann et al 2009) would be sufficient to form a stable global climate coalition for low levels of climate ambitions (e.g. global carbon price of USD 12.5 per tCO2e) or low (assumed) trade elasticities. The level of trade tariffs to maintain global co-operation would need to increase for higher trade elasticities (e.g. to 4%, (Lessmann et al 2009)) and higher mitigation ambition (e.g. 3% for USD 25 per tCO₂e, (Nordhaus 2015)). However, for higher global carbon prices (USD 50 and USD 100 per tCO2e), trade tariffs of even 10% would not be sufficient to constitute a stable global climate coalition. Yet, trade tariffs would still trigger participation of some regions (Nordhaus 2015).

their ambition on climate action, and thus facilitate countries' efforts to collectively meet the goals of the Paris Agreement. Quantifying the benefits of international co-ordination on pricing of CO_2 emissions and the distribution of these benefits across country groupings could help policy makers to make better-informed decisions about the implications of and potential forms for international co-ordination. Such forms could include harmonisation of carbon prices (e.g. through global or regional linking of carbon markets), extending coverage of pricing schemes, phasing out FFS, developing international sectoral agreements and co-ordination mechanisms to mitigate carbon leakage.

Our review shows that all forms of international co-operation on carbon pricing could deliver benefits which include economic benefits (e.g. lower mitigation costs) and environmental benefits (e.g. reducing GHG emissions and carbon leakage). Increasing mitigation in low-cost regions and reducing mitigation in high-cost regions achieves a given aggregate emissions target at a lower cost. Benefits tend to be higher with broader participation of countries, broader coverage of emissions and sectors and more ambitious policy goals (e.g. with emission reduction targets that align with the temperature goals of the Paris Agreement).

Yet, the economic benefits of international cooperation are likely to vary across countries and regions. Most countries would have substantial economic benefits from co-operation because of savings in mitigation cost (for international emissions trading) or reduced energy prices (for multilateral FFS removal). Some forms of co-operation would be unambiguously beneficial for all co-operating countries (e.g. extending the coverage of pricing schemes towards non-CO2 GHGs, linkages between countries with relatively similar mitigation ambition and abatement costs). Other forms of co-operation (e.g. multilateral FFS removal) would not always generate economic benefits for all countries. Redistributing the economic savings from co-operation across countries (e.g. via carbon market transactions, or potentially direct monetary transfers or technology transfers) could ensure that co-operation provides economic benefits for all countries. However, this may be politically challenging. Reinvesting the economic gains from co-operation into raised climate ambition would reduce long-term climate risks for all countries. Table 3 summarizes the core quantitative results and main findings regarding the different forms of co-operation.

All studies show substantial variation of carbon prices that would be implied by each region unilaterally meeting its specific mitigation targets, indicating a large potential for cost savings from harmonising carbon prices. Using carbon markets to help countries meet the mitigation goals in their NDCs with

¹⁵ This literature is summarized by Al Khourdajie and Finus (2020).

Table 3. Main quantitative findings for each co-operation instrument.

Main results	Specific evidence
National carbon prices needed to unilaterally reach submitted NDCs vary greatly across countries	Average regional simulated carbon prices in 2030 necessary to reach initial NDCs vary between \$6/tCO ₂
leaving room for efficiency gains from international co-ordination on carbon pricing	in Russia and \$119/tCO ₂ in the EU.
Instead of unilateral carbon pricing, global carbon	Global abatement costs for reaching NDCs in 2030 can
pricing can significantly reduce the overall costs of reaching NDCs	on average be reduced by 64%. The implied average global costs savings are \$229 billion in 2030.
For stricter targets, cost savings through a globally	For the 2 °C target, global costs can be reduced by on
harmonized price increase in absolute but decrease in	average by 50% or \$391 in 2030 for a global carbon
relative terms	price compared to regional carbon prices.
benefit all regions	across all studies from global harmonization of carbon prices. Generally, but especially for regional
	harmonization, developed regions mostly gain. Espe-
	basically all analysed targets and scenarios, does not
	gain from joining a trading regime in many studies.
Extending the sectoral coverage of pricing schemes	The highest positive effects of sectoral harmonization
reduces aggregate abatement costs	are found for the electricity sector. Extensions of
	carbon pricing to smaller sectors like transport or cement have positive but much smaller effects
Allowing flexibility in whether to abate CO ₂ or other	On average abatement costs would be between 23%
non-CO ₂ GHGs reduces abatement costs	and 48% lower in 2030 and 40% lower in 2050 with
	multi-gas mitigation rather than CO ₂ -only mitigation.
Sectoral agreements can reduce negative	For the cement sector, one study finds that the decrease
competitiveness effects of the covered sectors and imply welfare gains for participating countries yet	of cement production in the EU relative to a no policy case is reduced by around 36% through a joint ETS
emission reductions are limited and policy scenarios	with the cement sectors of China, Brazil and Mexico.
with a uniform global carbon price are preferable	
Globally phasing out FFS reduces GHG emissions and	Globally phasing out FFS reduces global CO ₂ -
increases global welfare	emissions of $1\%-4\%$ by 2030 relative to a no policy
	largest in fossil fuel exporting countries. Global welfare
	increases slightly.
Anti-leakage instruments are only an imperfect	BCA reduces leakage on average by 6% points
substitute for co-operation on carbon pricing	compared to scenarios where emission targets are
	reached without BCA and can reduce coalition
	countries wenate loss by up to 44%.

a uniform global carbon price has the potential to reduce global mitigation costs by on average 64%, translating into annual cost savings of on average USD 229 billion by 2030. The absolute, but not relative gains are higher for more ambitious mitigation targets. Regional emissions trading (e.g. through linking carbon markets) also brings benefits, albeit to a lower extent than global co-operation. Though there is no country or region that benefits in all studies from global harmonization of carbon prices, most developed countries/regions (e.g. Japan, EU, USA) would benefit economically and even more so from regional emissions trading, whereas this might not be the case for emerging economies (notably China). China could see a rise in domestic carbon prices under linked markets, which could negatively affect its international competitiveness vis-à-vis more developed and less carbon-intense economies. Similarly, extending the geographical scope of carbon markets by adding new countries would benefit most,

albeit not all countries in the absence of additional transfers.

Extending the coverage of carbon pricing schemes by including more sectors or non-CO₂ GHGs would deliver economic and environmental benefits, enabling countries to tap diverse sources of low-cost abatement options. International co-operation on reducing emissions in the power sector is estimated to have the largest potential for saving mitigation costs. Extending the coverage of (harmonised) carbon pricing beyond the power sector (e.g. to transport or industry) would further reduce aggregate mitigation costs, albeit to a lower extent. Extending the coverage of pricing schemes to non-CO2 GHGs would lead to average lower carbon prices by 23%-48% by 2030 compared to scenarios covering only CO₂ emissions. Sectoral agreements could potentially reduce sectorspecific GHG emissions and mitigate competitiveness concerns but are overall not efficient, though the evidence is scarce.

Global FFS removal by 2030 is estimated to reduce global CO₂ emissions by 1%-4% compared to BAU. Phasing out consumer FFS would increase domestic energy prices, reducing energy demand and emissions in the reforming countries, but may cause carbon leakage as a result of lower global energy prices, leading to increasing energy demand and emissions in other countries. Unilateral FFS removal would typically lead to economic gains for the reforming country due to more efficient resource allocation. Multilateral FFS reforms would also benefit most countries, notably energy-importing countries, compared to BAU, but would not be beneficial for some energy-exporting economies due to lower global energy prices. Globally, a multi-lateral FFS removal leads to slight welfare gains.

Co-ordinated implementation or increase of carbon pricing on a sub-global level (e.g. in form of a climate coalition or carbon club) would reduce carbon leakage within the coalition, but could increase carbon leakage outside. In the absence of multilateral agreements or co-ordinated efforts to reduce leakage, specific policy instruments (BCAs), carbon tax exemptions, allocation of free allowances) could reduce the risk of carbon leakage. Among those, BCA is expected to be most effective and would reduce leakage on average by 6% points and reduce welfare losses of the coalition by up to 44%. Yet, no instrument would be able to eliminate leakage entirely. BCA would bring economic benefits for coalition countries, but would, in general, disbenefit countries outside the coalition as it would transfer part of the mitigation effort to non-coalition countries whose exports essentially become taxed. Given the distributional implications, BCA could, in theory, provide incentives for non-coalition countries to join a climate coalition, but BCA's potential is expected to be limited.

The review is based on economic modelling studies, which are subject to some caveats. First, the studies and models reviewed here, including IAMs and CGE models, are stylised models that rely on a number of assumptions such as perfect rationality, information, and foresight of actors (e.g. households, firms) as well as perfect and complete markets. These assumptions are rarely observed in the real world. These assumptions lead to results where harmonized carbon pricing always leads to global economic benefits and thus, more broadly the global cost-reductions of the analysed scenarios should be interpreted as an upper bound of potential realworld effects. In fact, the estimated effects from modelling results far exceed those of empirical ex-post studies (see e.g. section 7). This can be explained by both the underlying assumptions of modelling studies and/or the discrepancy between actual and modelled policy variables (e.g. the level of carbon prices). Second, the results reported in the literature

neither capture all benefits associated with international co-operation nor all of its costs. Some models, notably IAMs assess the benefits associated with reduced long-term climate damages, but may not capture the full range of benefits from co-operation, including a reduced risk (and cost) of extreme events, or broader well-being benefits (reduced air pollution, reduced income inequality). Furthermore, most models quantify the short-term economic benefits, but inadequately evaluate the economic dynamics over the long-term. Regarding costs of co-operation, modelling studies typically do not account for the costs of setting up and maintaining co-operation, for harmonizing policies across nations or for monitoring cross-national carbon pricing schemes. The failure to capture the full costs is most pertinent in the most commonly discussed option for international co-operation i.e. international emission trading systems, which brings economic gains, but also results in on average lower international carbon prices in the absence of these costs. These low carbon price estimates that models report without fully capturing the full costs of establishing and maintain an international emissions trading system should not be misunderstood as if the mitigation costs are low. Such a misinterpretation may deter economic transformation and investments in innovation that would be needed to enable deep decarbonisation to reach netzero emissions by mid-century.

Lastly, overall our paper focuses on the quantitative results of the identified studies. Each of the proposed types of co-ordination could face challenges which could be political (e.g. domestic barriers to carbon pricing and FFS reforms; international burden sharing rules), practical (e.g. measuring emissions for different sectors) or legal (e.g. compatibility with international trade laws) that may impede implementation of carbon pricing. Also, implementing co-ordination mechanisms would require high levels of trust between the participating jurisdictions. However, the discussion of these challenges is beyond the scope of this paper.

Given these limitations, the reviewed studies nevertheless provide information about the potential reductions in economic abatement costs and/or additional emission reductions through the analysed cooperation scenarios. Even though our paper shows that the literature provides already many insights about the potential gains from international cooperation on carbon pricing and related climate policies, we also identified some problems, gaps and avenues for future research. First of all, it is often challenging to compare different studies due to different regional aggregations, target years, policy stringency, specific scenarios and reported variables and results. For this reason, multi-model studies within a harmonized setting and with harmonized reporting are especially helpful to identify the range

of results. The same is true for quantitative metaanalyses. These studies help at the same time to better understand the drivers of results, which even the multi-model studies mostly only touch upon without really explaining what is driving model results. More meta-analyses on issues where already sufficiently many studies exist (e.g. a linking of an EU and Chinese ETS or the gains from moving from unilateral do global carbon pricing under the Paris Agreement) could help to derive robust quantitative results and to understand what factors are driving them. Furthermore, some issues like a sectoral extension of carbon pricing, sectoral agreements as well as FFS reforms have received relatively little attention compared to the classical comparison of unilateral versus global carbon pricing, even though these topics might be of great political interest and practical relevance. Finally, another avenue for future work is to relax the neoclassical assumptions of models (e.g. perfect market, fully rational actors and more linking of economic models) with climate models to include feedback effects.

Data availability statement

The data that support the findings of this study are available upon reasonable request from the authors.

Appendix

Structure, metrics and caveats of economic models

Researchers use economic models, including CGE models and IAMs, to assess the effects of climate policy and international co-operation ex-ante. Economic models are a representation of the global economy, covering (representative) households and firms in different sectors (usually 2–15, but also up to 60) and different world regions (usually 5–20) that are connected through international markets (trade, capital). The time horizon ranges from 2030 or 2050 (CGEs) to as long as 2100 and beyond (mostly IAMs). Economic models require a number of input parameters and assumptions that determine the outputs as a result of the interplay of different systems.

Studies in this survey make use of multiple metrics on the (economic) effects of climate policies. All metrics are usually reported against a BAU scenario. While the climate policy's effect on emissions is straightforward and reported as reduced CO_2 or GHG emissions, different mitigation cost metrics exist (Paltsev and Capros 2013).

- Carbon price represents the marginal cost of an extra unit of emission reductions. Hence, this metric can be interpreted as mitigation effort, but not necessarily as the total cost of a policy.
- Loss in GDP represents the macroeconomic costs.

• Loss in welfare usually measures the amount of additional income needed for consumers to compensate for the consumption losses from a policy.

Two major channels can explain differences in the results from economic models across studies (Springer 2003). First, researchers may use different input parameters for BAU projections, including GDP, population, technological progress, etc. Second, results are usually sensitive to the choice of specific model parameters such as production elasticities. Hence, sound research needs to transparently display the assumptions regarding the input and model parameters while checking the robustness of the results for alternative parameter choices.

ORCID iD

Sneha Thube () https://orcid.org/0000-0002-9654-0102

References

- Abrell J 2010 Regulating CO₂ emissions of transportation in Europe. A CGE-analysis using market-based instruments *Transp. Res.* D **15** 235–9
- Akimoto K, Sano F, Oda J, Homma T, Uk R and Tomoda T 2008 Global emission reductions through a sectoral intensity target scheme *Clim. Policy* **8** S46–S59
- Akimoto K, Sano F and Tehrani B S 2017 The analyses on the economic costs for achieving the nationally determined contributions and the expected global emission pathways *Evol. Inst. Econ. Rev.* 14 193–206
- Al Khourdajie A and Finus M 2020 Measures to enhance the effectiveness of international climate agreements. The case of border carbon adjustments *Eur. Econ. Rev.* **124** 103405
- Aldy J E, Pizer W A and Akimoto K 2016b Comparing emissions mitigation efforts across countries *Clim. Policy* **17** 501–15
- Aldy J *et al* 2016a Economic tools to promote transparency and comparability in the Paris Agreement *Nat. Clim. Change* 6 1000–4
- Alexeeva V and Anger N 2016 The globalization of the carbon market. Welfare and competitiveness effects of linking emissions trading schemes *Mitig. Adapt. Strateg. Glob. Change* **21** 905–30
- Antimiani A, Costantini V, Kuik O and Paglialunga E 2016 Mitigation of adverse effects on competitiveness and leakage of unilateral EU climate policy. An assessment of policy instruments *Ecol. Econ.* 128 246–59
- Baranzini A, Jeroen V D B, Carattini S, Howarth R B, Padilla E and Roca J 2017 Carbon pricing in climate policy. Seven reasons, complementary instruments, and political economy considerations *Wiley Interdiscip. Rev. Clim. Change* **8** e462
- Böhringer C, Balistreri E J and Rutherford T F 2012a The role of border carbon adjustment in unilateral climate policy.
 Overview of an energy modeling forum study (EMF 29) Energy Econ. 34 S97–S110
- Böhringer C, Carbone J C and Rutherford T F 2012b Unilateral climate policy design. Efficiency and equity implications of alternative instruments to reduce carbon leakage *Energy Econ.* 34 S208–S217
- Böhringer C, Carbone J C and Rutherford T F 2016 The strategic value of carbon tariffs *Am. Econ. J.* **8** 28–51
- Böhringer C, Carbone J C and Rutherford T F 2018 Embodied carbon tariffs *Scand. J. Econ.* **120** 183–210
- Böhringer C, Dijkstra B and Rosendahl K E 2014a Sectoral and regional expansion of emissions trading *Resour. Energy Econ.* 37 201–25

- Böhringer C, Fischer C and Rosendahl K E 2010 The global effects of subglobal climate policies *B.E. J. Econ. Anal. Policy* **10** 1–35
- Böhringer C, Fischer C and Rosendahl K E 2014b Cost-effective unilateral climate policy design. Size matters J. Environ. Econ. Manage. 67 318–39
- Böhringer C, Peterson S, Rutherford T F, Schneider J and Winkler M 2021a Climate policies after paris. pledge, trade and recycle: insights from the 36th energy modeling forum study (EMF36) *Energy Econ.* **103** 105471
- Böhringer C, Rosendahl K E and Storrøsten H B 2017 Robust policies to mitigate carbon leakage *J. Public Econ.* **149** 35–46
- Böhringer C, Rutherford T F and Tol R S J 2009 THE EU 20/20/2020 targets. An overview of the EMF22 assessment Energy Econ. 31 S268–S273
- Böhringer C, Schneider J and Springmann M 2021b Economic and environmental impacts of raising revenues for climate finance from public sources *Clim. Policy* **21** 546–62
- Branger F and Quirion P 2014 Would border carbon adjustments prevent carbon leakage and heavy industry competitiveness losses? Insights from a meta-analysis of recent economic studies *Ecol. Econ.* **99** 29–39
- Burniaux J-M and Chateau J 2014 Greenhouse gases mitigation potential and economic efficiency of phasing-out fossil fuel subsidies *Int. Econ.* **140** 71–88
- Burniaux J-M, Chateau J and Duval R 2013 Is there a case for carbon-based border tax adjustment? An applied general equilibrium analysis *Appl. Econ.* **45** 2231–40
- Cain M, Lynch J, Allen M R, Fuglestvedt J S, Frame D J and Macey A H 2019 Improved calculation of warming-equivalent emissions for short-lived climate pollutants *npj Clim. Atmos. Sci.* 2 1–7
- CAT 2021 Climate action tracker 2021 (available at: https:// climateactiontracker.org/) (Accessed 19 August 2021)
- Cembureau 2013 The role of cement in the 2050 low carbon economy
- Chepeliev M and Dominique V D M 2020 Global fossil-fuel subsidy reform and Paris Agreement *Energy Econ.* **85** 104598
- Cosbey A, Droege S, Fischer C and Munnings C 2019 Developing guidance for implementing border carbon adjustments. Lessons, cautions, and research needs from the literature *Rev. Environ. Econ. Policy* 13 3–22
- Dai H-C, Zhang H-B and Wang W-T 2017 The impacts of US withdrawal from the Paris Agreement on the carbon emission space and mitigation cost of China, EU, and Japan under the constraints of the global carbon emission space *Adv. Clim. Change Res.* **8** 226–34
- Dechezleprêtre A, Gennaioli C, Martin R, Muûls M and Stoerk T 2019 Searching for Carbon Leaks in Multinational Companies (London: Centre for Economic Performance, LSE)
- Dellink R, Jamet S, Chateau J and Duval R 2014 Towards global carbon pricing. Direct and indirect linking of carbon markets *OECD J. Econ. Stud.* **2013** 209–34
- ECF 2014 The impact of including the road transport sector in the EU ETS
- EIA 2017 (available at: https://www.eia.gov/pressroom/ presentations/mead_91417.pdf) (Accessed 9 June 2020)
- Ellis J, Nachtigall D and Venmans F 2019 Carbon pricing and competitiveness 10.1787/f79a75ab-en
- Flachsland C, Brunner S, Edenhofer O and Creutzig F 2011 Climate policies for road transport revisited (II). Closing the policy gap with cap-and-trade *Energy Policy* 39 2100–10
- Fujimori S *et al* 2016 Will international emissions trading help achieve the objectives of the Paris Agreement? *Environ. Res. Lett.* **11** 104001
- G-20 2009 Leaders' statement (available at: https://g20.org/wpcontent/uploads/2014/12/Pittsburgh_Declaration_0.pdf (Accessed 11 March 2021))
- Gavard C, Winchester N, Jacoby H and Paltsev S 2011 Sectoral trading between the EU ETS and emerging countries *MIT JPSPGC Report, Appendix A* 193 vol 193

- Gavard C, Winchester N and Paltsev S 2016 Limited trading of emissions permits as a climate cooperation mechanism? US–China and EU–China examples *Energy Econ.* 58 95–104
- Ghosh M, Luo D, Siddiqui M S and Zhu Y 2012 Border tax adjustments in the climate policy context. CO₂ versus broad-based GHG emission targeting *Energy Econ*. 34 S154–S167
- Green J F 2021 Does carbon pricing reduce emissions? A review of ex-post analyses *Environ. Res. Lett.* **16** 043004
- Heinrichs H, Jochem P and Fichtner W 2014 Including road transport in the EU ETS (European Emissions Trading System). A model-based analysis of the German electricity and transport sector *Energy* 69 708–20
- Hof A F, Den Elzen M G, Admiraal A, Roelfsema M, Gernaat D E and van Vuuren D P 2017 Global and regional abatement costs of Nationally Determined Contributions (NDCs) and of enhanced action to levels well below 2 °C and 1.5 °C *Environ. Sci. Policy* **71** 30–40
- Höhne N *et al* 2021 Wave of net zero emission targets opens window to meeting the paris agreement *Nat. Clim. Chang.* 11 820–22
- Hübler M, Voigt S and Löschel A 2014 Designing an emissions trading scheme for China. An up-to-date climate policy assessment *Energy Policy* **75** 57–72
- ICAO 2020 (available at: https://www.icao.int/Newsroom/Pages/ ICAO-Council-agrees-to-thesafeguard-adjustment-for-CORSIA-in-light-of-COVID19-pandemic.aspx) (Accessed 3 August 2020)
- ICAP 2019 (available at: https://icapcarbonaction.com/en/ ?option=com_attach&task=download&id=625) (Accessed 9 June 2020)
- OECD/IEA 2019 Update on Recent Progress in Reform of Inefficient Fossil-Fuel Subsidies that Encourage Wasteful Consumption OECD/IEA
- IETA 2019 The Economic Potential of Article 6 of the Paris Agreement and Implementation Challenges IETA, University of Maryland and CPLC
- IISD 2018 (available at: https://www.iisd.org/publications/ advancing-linked-carbon-pricing-instruments-lessonsgoverning-carbon-pricing-clubs-non) (Accessed 9 June 2020)
- IMO 2018 Initial IMO strategy on reduction of GHG emissions from ships and existing IMO activity related to reducing GHG emissions in the shipping sector
- IPCC 2014 Climate change 2014 Synthesis report Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change ed IPCC R K Pachauri and L A Meyer (Geneva: IPCC) p 151
- Jewell J *et al* 2018 Limited emission reductions from fuel subsidy removal except in energy-exporting regions *Nature* **554** 229–33
- Kuik O, Brander L and Tol R S J 2009 Marginal abatement costs of greenhouse gas emissions. A meta-analysis *Energy Policy* 37 1395–403
- Larch M and Wanner J 2017 Carbon tariffs. An analysis of the trade, welfare, and emission effects *J. Int. Econ.* **109** 195–213
- Lessmann K, Marschinski R and Edenhofer O 2009 The effects of tariffs on coalition formation in a dynamic global warming game *Econ. Model* **26** 641–9
- Li M, Weng Y and Duan M 2019 Emissions, energy and economic impacts of linking China's national ETS with the EU ETS *Appl. Energy* **235** 1235–44
- Liu W, McKibbin W, Morris A C and Wilcoxen P J 2020 Global economic and environmental outcomes of the Paris agreement *Energy Economics* **90** 104838
- Liu Y and Wei T 2016 Linking the emissions trading schemes of Europe and China—combining climate and energy policy instruments *Mitig. Adapt. Strateg. Glob. Change* 21 135–51
- Magné B, Chateau J and Dellink R 2014 Global implications of joint fossil fuel subsidy reform and nuclear phase-out. An economic analysis Clim. Change 123 677–90

Meunier G and Ponssard J-P 2012 A sectoral approach balancing global efficiency and equity Environ. Resour. Econ. 53 533–52

Monjon S and Quirion P 2011 Addressing leakage in the EU ETS. Border adjustment or output-based allocation? *Ecol. Econ.* **70** 1957–71

- Mu Y, Evans S, Wang C and Cai W 2018 How will sectoral coverage affect the efficiency of an emissions trading system? A CGE-based case study of China *Appl. Energy* 227 403–14
- Nachtigall D 2019 Improving economic efficiency and climate mitigation outcomes through international co-ordination on carbon pricing 10.1787/0ff894af-en

Naegele H and Zaklan A 2019 Does the EU ETS cause carbon leakage in European manufacturing? J. Environ. Econ. Manage. 93 125–47

- Nong D and Siriwardana M 2018 The most advantageous partners for Australia to bilaterally link its emissions trading scheme Int. J. Glob. Warming 15 371–91
- Nordhaus W 2015 Climate clubs. Overcoming free-riding in international climate policy *Am. Econ. Rev.* **105** 1339–70 OECD 2019 Accelerating climate action *Refocusing Policies*

through a Well-being Lens (Paris: OECD Publishing) p 203 OECD 2021a Effective carbon rates 2021 (available at: www.oecd-

ilibrary.org/content/publication/0e8e24f5-en (Accessed/Retrieved 31 August 2021)) OECD 2021b OECD companion to the inventory of support

OECD 2021b OECD companion to the inventory of support measures for fossil fuels 2021 (available at: www.oecdilibrary.org/content/publication/e670c620-en (Accessed/Retrieved 31 August 2021))

Paltsev S and Capros P 2013 Cost concepts for climate change mitigation *Clim. Change Econ.* 4 1340003

- Peterson S and Weitzel M 2015 Reaching a climate agreement. Compensating for energy market effects of climate policy *Clim. Policy* 16 993–1010
- Qi T and Weng Y 2016 Economic impacts of an international carbon market in achieving the INDC targets *Energy* **109** 886–93

Schwanitz V J, Piontek F, Bertram C and Luderer G 2014 Long-term climate policy implications of phasing out fossil fuel subsidies *Energy Policy* **67** 882–94

Söder M, Thube S D and Winkler M 2019 The European emission trading system and renewable electricity: using the G/TAP power database to analyze the role of carbon prices on the development of renewables in the EU 22nd Annual Conference on Global Economic Analysis (Warsaw, Poland)

Springer U 2003 The market for tradable GHG permits under the Kyoto protocol. A survey of model studies *Energy Econ.* **25** 527–51

Stiglitz J E 2019 Addressing climate change through price and non-price interventions *Eur. Econ. Rev.* **119** 594–612

Tvinnereim E and Mehling M 2018 Carbon pricing and deep decarbonisation *Energy Policy* **121** 185–9

- UNEP 2019 Emissions gap report 2019 Executive Summary (Nairobi: United Nations Environment Programme)
- UNFCCC 2015 Adoption of the Paris Agreement—Paris Agreement (text English)
- Vandyck T, Keramidas K, Saveyn B, Kitous A and Vrontisi Z 2016 A global stocktake of the Paris pledges. Implications for energy systems and economy *Glob. Environ. Change* 41 46–63
- Venmans F, Ellis J and Nachtigall D 2020 Carbon pricing and competitiveness. Are they at odds? *Clim. Policy* 20 1070–91

Voigt S, Alexeeva-Talebi V and Löschel A 2012 Macroeconomic impacts of sectoral approaches. The role of the cement sector in China, Mexico and Brazil EcoMod2012 Eco. Mod.

Vrontisi Z *et al* 2018 Enhancing global climate policy ambition towards a 1.5 $^{\circ}$ C stabilization. A short-term multi-model assessment *Environ. Res. Lett.* **13** 044039

Walley D 2012 How large are the impacts of carbon motivated border tax adjustments? *Clim. Change Econ.* **03** 1250001

Wei Y-M et al 2018 An integrated assessment of INDCs under shared socioeconomic pathways. An implementation of C3IAM Nat. Hazards 92 585–618

Weitzel M, Hübler M and Peterson S 2012 Fair, optimal or detrimental? Environmental vs strategic use of border carbon adjustment *Energy Econ.* 34 S198–S207

Weyant J P, Francisco C and Blanford G J 2006 Overview of EMF-21. Multigas mitigation and climate policy *Energy J*. SI2006 1–32

World Bank 2021 State and trends of carbon pricing 2021: the World Bank (available at: https://openknowledge.world bank.org/handle/10986/35620 (Accessed 31 August 2021))

WSA 2019 Steel's contribution to a low carbon future and climate resilient societies